Author's response to reviews

Title: Analyzing International Clinical Education Practices for Canadian Rehabilitation Students

Authors:

Puja Ahluwalia (puja.ahluwalia@alum.utoronto.ca)
Deb Cameron (deb.cameron@utoronto.ca)
Lynn Cockburn (l.cockburn@utoronto.ca)
Lynn Ellwood (lynn.ellwood@utoronto.ca)
Brenda Mori (brenda.mori@utoronto.ca)
Stephanie A Nixon (stephanie.nixon@utoronto.ca)

Version: 3
Date: 5 August 2014

Author's response to reviews: see over
Dear Editorial Team at BMC Medical Education,

Thank you for the constructive reviews of our manuscript. We have ensured that the qualitative component of our study adheres to the RATS guidelines for reporting qualitative studies and have incorporated a statement to that effect in the Methods section of the manuscript.

Sincerely,
Puja Ahluwalia